Author’s response to reviews

Title: Rapid Development of HIV Elite Control in a Patient with Acute Infection

Authors:
Deirdre Morley (deemorley@gmail.com)
John Lambert (jlambert@mater.ie)
Louise Hogan (lou.hogan3@gmail.com)
Cillian De Gascun (cillian.degascun@ucd.ie)
Niamh Redmond (n.redmond@ucd.ie)
Rachel Rutishauser (Rachel.Rutishauser@ucsf.edu)
Cassandra Thanh (Cassandra.thanh@ucsf.edu)
Erica Gibson (gibsonerica14@gmail.com)
Kristen Hobbs (Kristen.Hobbs@ucsf.edu)
Sonia Bakkour (SBakkour@vitalant.org)
Michael Busch (MBusch@vitalant.org)
Jeremy Farrell (jfarrell@mater.ie)
Padraig McGetrick (padraigmgetrick@gmail.com)
Timothy Henrich (Timothy.Henrich@ucsf.edu)

Version: 3 Date: 05 Aug 2019

Author’s response to reviews:

We thank the reviewers for their insightful and positive comments on the manuscript. We have addressed all the comments as below which we feel significantly strengthens the paper.

1. In the Availability of data and materials section please clearly state who should be contacted if
someone wants to request the data.

The corresponding author Deirdre Morley should be contacted. (manuscript amended)

2. In the “Funding” section of your declarations, please clarify the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Divisional Discretionary Funds from UCSF. This research was supported by the National Institute of Health, National Institute of Allergy and Infectious Diseases grants AI141003 and UM1AI126611. The content is solely the responsibility of the authors and does not necessarily represent the official views of the National Institutes of Health. (manuscript amended)

3. Please remove the institutional information of authors from the "Author Contributions" section of the declarations and instead provide it in a new section entitled "Authors' information" after the "Acknowledgements".

Manuscript amended